Author’s response to reviews

Title: IT-adoption of clinical information systems in Austrian and German hospitals: results of a comparative survey with a Focus on nursing

Authors:

  Ursula Hübner (u.huebner@fh-osnabrueck.de)
  Elske Ammenwerth (elske.ammenwerth@umit.at)
  Daniel Flemming (d.flemming@fh-osnabrueck.de)
  Christine Schaubmayr (christine.schaubmayr@tilak.at)
  Björn Sellemann (b.sellemann@fh-osnabrueck.de)

Version: 5 Date: 21 January 2010

Author’s response to reviews:

This is the final version of manuscript which includes all requested changes (see your eMail dated 20 January 2010)
in particular
we checked all URLs
we produced all figures again so that they occupy more space of the white sheet
we also changed the spelling of IT-xxx to IT xxx, e.g. IT-system to IT system.